Author's response to reviews

Title: Necrotizing fasciitis- a diagnostic dilemma: Two case reports

Authors:

Mitrakrishnan R Navinan (rayno.navinan@gmail.com)
Jevon Yudhishdran (yudhishdran@yahoo.com)
Thambyaiah Kandeepan (kathambyaiah@yahoo.com)
Aruna Kulatunga (kulatunga.aruna@yahoo.com)

Version: 3
Date: 17 April 2014

Author's response to reviews:

Ms Anna Kathrina Vidal
Journal Editorial Office
BioMed Central

Dear Madam,

I have made the amendments as requested in your email sent to me dated April 16th, 2014.

1. Figure legend section header added
2. Images containing any remaining patient identifiers have been whited out, please note MRI images contains data, but it is standard imaging data
3. All images have been cropped to maximize the image and remove space as much as possible while maintaining image ratio in length and width

I hope these changes are sufficient, and I have uploaded the documents and images using this instead of emailing back as requested

Thanking you,

Rayno Navinan